Author's response to reviews

Title: The implementation of the Medical Regulation Office and Mobile Emergency Attendance System and its impact on the gravity profile of non-traumatic afflictions treated in a University Hospital: a research study

Authors:
Sergio L B Lopes (siblopes@uol.com.br)
Jose Sebastiao DOS Santos (jsdsanto@fmrp.usp.br)
Sandro Scarpelini (sandro@fmrp.usp.br)

Version: 2 Date: 12 February 2007

Author's response to reviews:

The authors revised the paper as demanded by the reviewers, doing the necessary changes according to their suggestion.

The study "the implementation of Medical Regulation Office and Mobile Emergency Attendance System and it's impact on the gravity profile of non-traumatic affections treated in a University Hospital: a research " was motivated by the need to test the perception of the impact produced by implementation of the medical regulation (MR) and of the Mobile Emergency Attendance System (SAMU) in the organization of patient access to the University Hospital. It was demonstrated that this device, besides redirecting the cases less serious to other components of the attendance net, can increase the profile of gravity of the patients, fortifying the principles of equity in access to health resources. These process can induce important changes in attention given to emergencies and in the organization of the process of learning. The MR and SAMU represent an alternative, to be evaluated by the public systems, and as this study design has not been described previously, we judge that the work must be published in the high impact International Journal.